Author’s response to reviews

Title: S100A4 is elevated in axial spondyloarthritis: a potential link to disease severity

Authors:
Barbora Šumová (sumova@revma.cz)
Lucie Andrés Cerezo (andres-cerezo@revma.cz)
Hana Hulejová (hulejova@revma.cz)
Klára Prajzlerová (prajzlerova@revma.cz)
Michal Tomčík (tomcik@revma.cz)
Kristýna Bubová (bubova@revma.cz)
Jan Štěpán (stepan@revma.cz)
Mária Filková (filkova@revma.cz)
Tereza Kropáčková (kropackova@revma.cz)
Mariam Grigorian (grigorian@sund.ku.dk)
Karel Pavelka (pavelka@revma.cz)
Jiří Vencovský (vencovsky@revma.cz)
Ladislav Senolt (senolt@revma.cz)

Version: 2 Date: 09 Dec 2019

Author’s response to reviews:

December 9th, 2019

Dear editorial office,

We have implemented the minor changes as suggested by the reviewer 1.

We believe that we were able to modify the paper accordingly that it could be reconsidered for publication.

Sincerely Yours

Lucie Andrés Cerezo